\documentclass{article}

\usepackage[round]{natbib}
\usepackage{hyperref}
\hypersetup{
  colorlinks = true, %Colours links instead of ugly boxes                     
  urlcolor   = blue, %Colour for external hyperlinks                          
  linkcolor  = blue, %Colour of internal links                                
  citecolor  = red   %Colour of citations                                        
}

\begin{document}

Papers cited: 
\begin{enumerate}
 \item \cite{bendoni_et_al_OM23} 
 \item \cite{sciascia_et_al_ICESJ22}    
 \item \cite{reyes_et_al_OS22}  
 \item \cite{lorente_et_al_OS22}  
 \item \cite{vonschuckmann_et_al_JOPOCEAN20}  
 \item \cite{mazoyer_et_al_ECSS20}  
 \item \cite{enrile_et_al_PLOSONE19}
 \item \cite{sciascia_et_al_DSRI19}
 \item \cite{fraser_et_al_JGR18}
 \item \cite{sciascia_et_al_OS18}
 \item \cite{berta_et_al_OS18}
 \item \cite{enrile_et_al_CSR18}
 \item \cite{almansi_et_al_JPO17}
 \item \cite{gelderloos_et_al_JPO17}
 \item \cite{koszalka_et_al_OM17}
 \item \cite{suaria_et_al_SREP16}
 \item \cite{mariano_et_al_JGR16}
 \item \cite{carlson_et_al_CSR16}
 \item \cite{bellomo_et_al_JOPOCEAN16}
 \item \cite{berta_et_al_JTECH15}
 \item \cite{magaldi_haine_DSRI15}
 \item \cite{aliani_et_al_RL15}
 \item \cite{berta_et_al_PO14}
 \item \cite{vonappen_et_al_DSRI14}
 \item \cite{koszalka_et_al_JPO13}
 \item \cite{magaldi_et_al_JPO11}
 \item \cite{degaetano_et_al_AQRES11}
 \item \cite{magaldi_et_al_OCDYN10}
 \item \cite{poje_et_al_OCMOD10}
 \item \cite{mercantini_et_al_OCMOD10}
 \item \cite{magaldi_et_al_OCMOD08}
 \item \cite{degaetano_et_al_AQRES08}
 \item \cite{doglioli_et_al_JGR04}
 \item \cite{doglioli_et_al_AQUA04}
\end{enumerate}

Datasets cited:
\begin{enumerate}
 \item \cite{berta_et_al_SEANOEdrifterDDR20}   
 \item \cite{berta_et_al_SEANOEdrifterSP} 
 \item \cite{magaldi_et_al_ZENODOtrieste}
\end{enumerate}

\bibliographystyle{plainnat}
\bibliography{MagaldiBib}

\end{document}
